Author's response to reviews

Title: CT based three dimensional dose-volume evaluations for high-dose rate intracavitary brachytherapy for cervical cancer

Authors:

Naoya MD, PhD Murakami (ore-murakami@hotmail.co.jp)
Takahiro MD, PhD Kasamatsu (takasama@ncc.go.jp)
Akihisa MSc Wakita (awakita@ncc.go.jp)
Satoshi MSc Nakamura (satonaka@ncc.go.jp)
Hiroyuki PhD Okaoto (hiokamot@ncc.go.jp)
Madoka MD, PhD Morota (madmorot@ncc.go.jp)
Minako MD, PhD Sumi (msumi@ncc.go.jp)
Koji MD Inaba (koinaba@ncc.go.jp)
Yoshinori MD Ito (yito@ncc.go.jp)
Jun MD, PhD Itami (jitami@ncc.go.jp)

Version: 2
Date: 18 April 2014

Author's response to reviews: see over
April 18th, 2014

Prof. Dafne Solera, M.D.
----- Executive Editor, BMC Cancer
Middlesex House 34-42 Cleveland Street
London, London W1T 4LB
United Kingdom

Dear Professor Solera

Thank you for taking our article into possible future publication for your journal. Please find revised version of our article and attached replies for reviewer’s comment. We hope you accept our revised version of our manuscript.

Sincerely yours,

Naoya Murakami M.D., Ph.D.
Department of Radiation Oncology, National Cancer Center Hospital, Tokyo, Japan
104-0045, 5-1-1 Tsukiji, Chuo-ku, Tokyo 104-0045, JAPAN
Phone: +[81]-3-3542-2511
FAX: +[81]-3-3545-3567
E-mail: ore-murakami@hotmail.co.jp, namuraka@ncc.go.jp